**Author’s response to reviews**

**Title:** Perinatal depressive symptoms among low-income South African women at risk of depression: trajectories and predictors

**Authors:**

Emily Garman (emily.garman@uct.ac.za)

Marguerite Schneider (marguerite.schneider@uct.ac.za)

Crick Lund (crick.lund@uct.ac.za)

**Version:** 2  
**Date:** 06 Jun 2019

**Author’s response to reviews:**

Thank you for providing feedback on our manuscript. As per your suggestion, we have now uploaded a clean version of our manuscript (with no track changes). We have checked that all acronyms are defined in the manuscript, and an 'Abbreviations' section was included in the manuscript before the 'Declarations' section. We have also amended the grammatical error spotted in the Discussion section.